Title: The use of Intra-antral application of antifungal agent in the treatment of Recurrent Maxillary Fungal Rhinosinusitis: Case report

Authors:

Adekunle D Dunmade (dumkun@yahoo.com)
Olushola A Afolabi (droaafolabi@yahoo.com)
Biodun S Alabi (alabibs@yahoo.com)
Segun Segun-Busari (segunsegunbusari@yahoo.com)
Olubisi A Koledoye (olubisikoledoye@yahoo.com)

Version: 6 Date: 29 May 2012

Author’s response to reviews: see over
The Editor,

Journal of Medical Case Report,

Dear Sir,

RE- THE USE OF INTRA-ANTRAL APPLICATION OF ANTIFUNGAL AGENT IN THE TREATMENT OF RECURRENT MAXILLARY FUNGAL RHINOSINUSITIS: CASE REPORT.

We wish to submit the corrected form of the above titled manuscript to your highly esteemed journal. We the authors of the above titled article have gone through the reviewer’s comment and have done the following corrections.

We have changed the article type to case report on the manuscript type

We have also included the tribe in the article and the abstract as West African Yoruba

The images have also been removed sir and the texts that could reveal patient identity have been removed

Thanks

Yours faithfully,

Afolabi O.A

Correspondence author